Author’s response to reviews

Title: Spatial clustering of fatal, and non-fatal, suicide in New South Wales, Australia: Implications for evidence-based prevention

Authors:

Michelle Torok (m.torok@unsw.edu.au)
Paul Konings (paul.konings@anu.edu.au)
Philip Batterham (philip.batterham@anu.edu.au)
Helen Christensen (h.christensen@unsw.edu.au)

Version: 3 Date: 14 Sep 2017

Author’s response to reviews:

Dear Mr Hickey,

Please see our response to the following comment:

1. Please state whether the maps depicted in your figures are your own or taken from another source.

All maps provided as part of the manuscript are our own, generated by Mr Paul Konings (second author) using the arcGIS platform, from the two data sets that we acquired and have been described within the manuscript (ABS mortality data, NSW Health Admitted Patient Data Collection).

We hope this response is satisfactory, and please do not hesitate to seek further clarification on any other issues that may arise.

Many thanks,

Michelle Torok.